Author's response to reviews

Title: N-acetyltransferase 8, a positional candidate for blood pressure and renal regulation: resequencing, association and in silico study

Authors:

Peeter Juhanson (peeter.juhanson@ut.ee)
Katrin Kepp (katike@ut.ee)
Elin Org (elin@ut.ee)
Gudrun Veldre (gveldre@ut.ee)
Piret Kelgo (piret.kelgo@hot.ee)
Mai Rosenberg (mai.roenberg@kliinikum.ee)
Margus Viigimaa (margus.viigimaa@regionaalhaigla.ee)
Maris Laan (maris@ebc.ee)

Version: 3 Date: 1 March 2008

Author's response to reviews: see over
Dear BMC Medical Genetics Editors,

Dear reviewers,

Thank you for reading and additional comments for the manuscript submitted to BMC Medical Genetics by Juhanson et al “N-acetyltransferase 8, a positional candidate for blood pressure and renal regulation: resequencing, association and in silico study”.

We have improved the manuscript according to the further suggestions by the reviewer no 2 (Dr. Pankow). The reviewer no 3 (Dr. Munroe) was satisfied with the previous edits and advised acceptance without revision.

Materials and Methods:
1. Data about the participation rate in sample recruitment has been added.
2. Details about the application of the two devices for blood pressure measurements have been supplemented.
3. The estimation of measurement uncertainty of the serum biomarkers of has been described.

Discussion: the wording of two sentences has been improved based on suggestions (4) and (5).

The article text has been edited and typos corrected by an English philologist in the style of British English.

Yours sincerely,

Maris Laan,
Corresponding author
Research professor
Institute of Molecular and Cell Biology, University of Tartu